Author's response to reviews

Title: Recurrent spontaneous pneumothorax during pregnancy: a case report.

Authors:

Preeti Jain (preetijain@hotmail.co.uk)
Kavita Goswami (gynaedocto@aol.com)

Version: 2 Date: 31 May 2008

Author's response to reviews:

Dear Editor,

Thanks for your letter regarding my manuscript.

I have made the suggested changes, which are as follows:

1. Rephrasing case report to case presentation
2. Ethic origin and other details of the patient
3. E-mail ID of the authors in the title page
4. Abbreviations
5. Acknowledgements

Thanking you.

Yours sincerely

Preeti Jain